Author's response to reviews

Title: Association between paraoxonase gene and stroke in the Han Chinese Population

Authors:

Guojun Zhang (zguoj1018@yahoo.com.cn)
Wenjin Li (wenjinliwlj@gmail.com)
Zhiqiang Li (lizqsitu@gmail.com)
Hong Lv (lvhonghong@yahoo.com.cn)
Yonghong Ren (yhren@capitalbio.com)
Ruimin Ma (ma.ruimin@yahoo.com.cn)
Xiaohong Li (xhli@capitalbio.com)
Xixiong Kang (kangxx@VIP.sina.com)
Yongyong Shi (shiyongyong@gmail.com)
Yimin Sun (ymsun@capitalbio.com)

Version: 8 Date: 18 December 2012

Author's response to reviews: see over
Dear Editors,

Thank you for your careful review of our manuscript entitled as “Association between Paraoxonase gene and stroke in the Han Chinese Population”. Those comments are very valuable and helpful for improving our manuscript. We have carefully revised the manuscript according to the comments. We hope this revision can be accepted by your journal.

Sincerely yours,

Wenjin Li, wenjinliwl@gmail.com

Corresponding author:

Name: Yimin Sun, E-mail: ymsun@capitalbio.com;

Name: Yongyong Shi, E-mail: shiyongyong@gmail.com.

The following is a point-to-point response to the comment.

1. We advise you to seek the assistance of a fluent English speaking colleague, or to have a professional editing service correct your language. Please ensure that particular attention is paid to the abstract.

**Answer:** Thank you very much. We did seek the help of a professional editing service to correct the language and carefully revised the abstract again. We hope the language is acceptable now.